Author’s response to reviews

Title: Subclinical Hypothyroidism Would Not Lead To Female Sexual Dysfunction in Chinese Women

Authors:

Han Luo (byrant_luohan@live.cn)
Hongliu Yang (saturday_me@msn.cn)
Wanjun Zhao (524675158@qq.com)
Qianqian Han (1173825864@qq.com)
Li Zeng (32471488@qq.com)
Huairong Tang (HuairongTang@163.com)
Jingqiang Zhu (zjq-wkys@163.com)

Version: 4 Date: 21 Sep 2017

Author’s response to reviews:

Dear editor,

I have revised the manuscript as much as i can.

However, I am kind of confused by your similarity index. The highlight part also includes large part of authors' names, affiliation, and various terminology. So it is really hard to avoid these words to be overlapped, because it is fixed format.

And I think the most overlapped part is from method part. Because this manuscript is from the same project as another paper, so it is reasonable to have the similar methodology.

I appreciate you reconsider the value of the similarity index.

Thanks

Jingqiang Zhu